Author's response to reviews

Title: Maternal perspectives on the use of probiotics in infants: a cross sectional survey

Authors:

Sarah L Bridgman (sarah.bridgman@ualberta.ca)
Meghan B Azad (meghan.azad@ualberta.ca)
Catherine J Field (cjfield@ualberta.ca)
Nicole Letourneau (nicole.letourneau@ucalgary.ca)
David W Johnston (davidw.johnston@albertahealthservices.ca)
Bonnie J Kaplan (bonnie.kaplan@albertahealthservices.ca)
Anita L Kozyrskyj (kozyrsky@ualberta.ca)

Version: 2 Date: 20 July 2014

Author's response to reviews: see over
Re: Manuscript Submission - revisions

Dear Dr. Rowles,

As requested in your email dated July 18th, 2014 I have made the following revisions to the submission of our manuscript:

1. Addition of line number in the main manuscript document
2. Provision of a copy of the questionnaire used in our study

We look forward to the comments from the peer review process of our manuscript.

Sincerely,

Anita Kozyrskyj, PhD
Professor
Department of Pediatrics, University of Alberta